Dear reviewers,

Thank you very much for considering accepting our case. We have read the peer-review report and the comment from the company editor-in-chief carefully. Here are our answers.

The comments from the reviewers:

1. The title and abstract of the manuscript are consistent with the content of the manuscript. The topic is within the scope of the WJCC. The figures are highly suggestive of the paper contents. The manuscript is very well organized and presented. The authors provided the CARE Checklist–2016 and written informed consent. The authors found an extremely rare case and have concisely presented the clinical and therapeutic background known to date. This clinical case is an extremely interesting and useful one for fellow clinicians because of the reduced number of cases reported so far and variety of paraclinical investigations used.

   Answer: Thank you for your valuable comment.

2. very well written informative article i accept

   Answer: Thank you for your valuable comment.

The comment from the Company Editor-in-Chief:
1. I have reviewed the Peer-Review Report, full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

Answer: Thank you for your valuable comment. We have provided the original figure documents using PowerPoint, which can be reprocessed by the editor. We have uploaded our funding agency copy, too.

Thank you again for the comments from the reviewers and the company editor-in-chief. I am looking forward to hearing from you.

Sincerely yours,

Shaohua Qu

Oct 30th, 2021